Dear Dr Peiro Lo Monaco,

Thank you for your email and the opportunity to revise our paper on "Liver Shape Analysis using Statistical Parametric Maps at Population Scale". We have provided a clear statement in the Funding section as per guidance. We also appreciate your comments, and the suggestions offered by the reviewers have been very useful.

I have included the Reviewer's comments and our responses immediately after this letter. All modifications in the manuscript have been highlighted in yellow.

Yours sincerely,

Marjola Thanaj

## Reviewer #1

This authors provided responsive edits to the manuscript to address my statistical concerns. While the authors refused to recognize or discuss the potential impact of segmentation and registrations errors on their approach, this reviewer believes that this point should not be a barrier towards publication of the authors' undoubtedly impressive work. However, this reviewer would caution the authors going forward that although the S2S distances may be relevant at the population scale where the confidence interval around an average trend may be narrow enough to detect statistical significance, the prediction interval for an individual patient, which statistically does not benefit from the same large cohort size, may be worrisome due to high sample variances arising from error sources such as the aforementioned uncertainties in segmentation and registration. As this work continues to develop, it will remain to be seen how well the S2S distances as potential biomarkers will perform in reproducibility and reliability studies that eventually will be necessary for the proposed ideas to be clinically recognized and adopted.

## Response:

 Thank you for your thoughtful comments and concerns regarding the potential impact of segmentation and registrations errors as well as the associated challenges when applied to an individual patient. As our work progresses, we will definitely assess and address your suggestions in our future work. Thank you for your valuable input and for considering our work for publication.